Author’s response to reviews

Title: The Pap smear screening as an occasion for smoking cessation and physical activity counselling: Baseline characteristics of women involved in the SPRINT randomized controlled trial

Authors:

Elisabetta Chellini (e.chellini@ispo.toscana.it)
Giuseppe Gorini (g.gorini@ispo.toscana.it)
Giulia Carreras (g.carreras@ispo.toscana.it)
Livia Giordano (livia.giordano@cpo.it)
Emanuela Anghinoni (emanuela.anghinoni@aslmn.it)
Anna Iossa (a.iossa@ispo.toscana.it)
Cristina Bellati (cristina.bellati@cpo.it)
Elisa Grechi (eli.gre@virgilio.it)
Alessandro Coppo (alessandro.coppo@cpo.it)
Talassi Fiorella (talassifiorella@aslmn.it)
Maria Rosa Giovacchini (mariarosa.giovacchini@asf.toscana.it)

Version: 4 Date: 22 November 2011

Author’s response to reviews:

Dear Editor,

reviewing the formatted version of the paper I discovered few spelling errors both in the text and in the figure. The revised files with the spelling errors highlighted will be soon uploaded.

Many thanks for your attention,

best regards

Elisabetta Chellini